Author's response to reviews

Title: PrognoScan: a new database for meta-analysis of the prognostic value of genes

Authors:

  Hideaki Mizuno (mizunohda@chugai-pharm.co.jp)
  Kunio Kitada (kitadakno@chugai-pharm.co.jp)
  Kenta Nakai (knakai@ims.u-tokyo.ac.jp)
  Akinori Sarai (sarai@bse.kyutech.ac.jp)

Version: 2 Date: 10 February 2009

Author's response to reviews:

We found several points which do not conform to the journal style, such as fonts and headings. We thus corrected them in the manuscript.